Author's response to reviews

Title: Bilateral Hilar Lymphadenopathy in a Young Female: Case Report Seema Varma, MD; Shilpi Gupta, MD; Raymond El-Souidie, MD; Meekoo Dhar, MD; Neville Mobarakai, MD Sanford R. Nalitt Institute of Cancer and Blood Related Diseases, Staten Island University Hospital, New York.

Authors:

Seema Varma MD (svarma@siuh.edu)
Shilpi Gupta MD (sgupta@siuh.edu)
Raymond El-Souidie MD (relsouidie@siuh.edu)
Meekoo Dhar MD (mdhar@siuh.edu)
Neville Mobarakai MD (nmobarakai@siuh.edu)

Version: 2 Date: 20 March 2007

Author's response to reviews:

This is an interesting case of a young African American female presenting with bilateral hilar and mediastinal lymphadenopathy and symptoms suggestive of sarcoidosis. This case asserts that Bronchiolitis Obliterans - Organizing Pneumonia can mimic presentation of various diseases, including sarcoid.